Author's response to reviews

Title: Inequalities in public water supply fluoridation in Brazil: an ecological study

Authors:

Marilisa CL Gabardo (marilisagabardo@e-odonto.com)
Wander J da Silva (wanderjose@e-odonto.com)
Marcia Olandoski (bio.estatistica@pucpr.br)
Simone T Moyses (simone.moyses@pucpr.br)
Samuel J Moyses (s.moyses@pucpr.br)

Version: 6 Date: 9 April 2008

Author's response to reviews: see over
April, 9th 2008.

MS: 8117245291467114

Inequalities in public water supply fluoridation in Brazil: an ecological study
Marilisa CL Gabardo, Wander J da Silva, Marcia Olandoski, Simone T Moyses and Samuel J Moyses

One more time we wish to express our thanks for the opportunity in publishing in BMC Oral Health. All the process in the construction of this paper with the useful suggestions from the reviewers was excellent.

Sincerely,

Marilisa CL Gabardo
Marcia Olandoski
Samuel J Moyses

Wander J da Silva
Simone T Moyses